MEG Management Committee meeting

21.02.2008, 13:00hrs, West Wing Lecture Theatre, CBU

Minutes
Present:  Friedemann Pulvermuller (Chair), Rik Henson, Yury Shtyrov, Rob Anthony, Olaf Hauk, William Marslen-Wilson, Clare Dine, Georgie Morrill (minutes)
1. Apologies for absences 

· Maarten Van Casteren (operating)
· Oleg Korzyukov (operating)

· Rhodri Cusack

· Gary Chandler

2. Matters arising 
YS informed the committee that he had been liaising with Neuromag about the recent issues with the lab.  WM advised that we should keep feeding back any new or unsolved problems to Neuromag.  FP and WM had also received an email asking for feedback regarding the Maxshield in order for them to improve the product.  As FP pointed out though, we have had too many problems with the lab in general to comment on this specific thing and would need more time to provide a more objective opinion.  WM suggested that YS draft a letter to Neuromag to explain this.
    Action:  YS draft letter to Neuromag.
3. Minutes from the last meeting
WM questioned whether we were charging partners for pilot slots.  RH confirmed that we are as this is meant to be a slot for a full recording session as a practice before the study begins.  The only problems we have had are when some researchers have used this time just to sort out their stimuli etc.  WM asked whether we could use Mimic to get round this somehow and RH agreed that this could be done and perhaps one of the testing rooms could be used to store the Mimic machine.  FP also pointed out that it would be good to get the EEG set up room closer to the lab.  Again, it was suggested that possibly one of the testing rooms could be used. WM suggested looking into the usage of these rooms.  YS pointed that setting up a MEG mimic machine would not be trivial as it also requires a mock of the complete data stream, and using the real acquisition electronics for this would be too risky for the recordings. It was agreed to continue exploring possible mimic options.
The minutes were approved.
4. Computing
RA reported that we now have 5 working machines in the User room which all have 2 20”screens each, have OpenGL graphics cards.  Yury brought up the fact that there are some issues with one of the machines which computing were not aware of, so Rob will look into this.  RA also informed the committee that there are 6 new machines on the central rack which also have OpenGL graphics cards and he and Dan Wakeman have successfully run a remote VNC session on them.  Both Rik and Yury have also used a machine and stated that they are quite fast.  Yury said that he gets a flicker on the screen of the desktop and RA will also be looking into this.  Yury thinks that the 24’’ screen is certainly usable, though not quite as good as having two 20” screens – especially when it comes to using MNE or Neuromag software where you often have five to six panels up at one time.  There is a possibility to increase the screen sizes in the future when the prices have decreased, because to get a higher resolution than a 24” screen would require moving to a 30” screen which are currently prohibitively expensive.  It will be discussed outside the meeting whether any individual needs a larger screen if frequently running remote VNC sessions.

All the data streams for the partners have been created so they should be able to access these now to retrieve data.  CD reported that Alex Clarke from CSL had problems saving the data.  All other CSL users haven’t had these issues and RA will fix this problem.
The new imaging disk has now been created specifically for MEG use.  It is about 12 TB so will have much more room for the data.  RA still needs to move the data across so this is ongoing.  WM suggested that we should enforce good ‘housekeeping’ anyway because eventually the 12TB will be used.
RA has ordered the new licenses for Neuromag software.  He could not obtain a permanent license for MCE so we have just a temporary one which will need to be renewed in 6 months time.  FP announced that he had issues using the Linux machines in the User room which RA will investigate.

RH informed the committee that in part of the tender with Partners, it states that we should provide the software that they will need to analyse their data.  RA gave CSL the details they needed to obtain this and RH said that they have already installed some of it.  RH brought up the issue of the cross-talk and fine calibration files which are required by Maxfilter but stored on our network.  RA suggested that we could have an area on the data gateway to store this which will mirror any updates and all partners obviously have access to.  A script will need to be run to do this.
RA confirmed to YS that when the computing system is down it impossible to record in the MEG lab because the project database is stored on the networked data space.  RA suggested a possible solution, which would be to store the database with project settings on the ftp gateway which moves the data onto the system for which you only need the host running to enable it to work.  It will then be possible to save the data on the central disk array. Although it is still possible that the ftp gateway may be down it is much less likely to happen than the data space. FP suggested that the data could be saved on the local acquisition computer but RA thinks this will be more difficult to back up and make secure. It was agreed to move the database to the ftp gateway.
WM queried if the visitor requests for partners were related to them getting accounts at the CBU.  RH announced that it was mainly to do with out of hours working, but maybe some could have accounts.  RA pointed out that this is not desirable as this would lead to security issues as then our Partners would be able to access our entire network. A discussion followed, and it was stated that we should avoid the situation when the CBU may become responsible for fixing any bugs or giving training, doing updates etc.    WM said that they will still ask for our advice etc, which everybody agreed was OK.
YS asked about moving a workstation into the EEG lab and RH pointed out that this would be helpful for preparing project settings, but OH queried the whether there is enough space. It was also queried whether a use of additional workstation in the lab for analysis could compromise the recordings. RA said that this was possible from a technical point of view and the committee decided that YS would organise moving one HP machine for a trial period. A set of monitors would be required for this which IT would need to provide.
Action:  RA to investigate screen flicker problem, data saving with Alex Clarke, Linux machine issues in User room, and storing of Maxfilter files so that Partners can access.  YS and RA to coordinate move of HP work station to MEG lab.
5. Recent lab issues with data acquisition/processing
YS reported that the main issues were noise, bad HPI fits, dropping channels and Neuromag are investigating this and liaising with YS.  RH remembers Danny mentioning lights effecting HPI.  YS said that this was initially investigated by YS and MvC and that there was noticeable ‘mains’ noise which stopped as soon as the lights went off.  WM queried the fact the lab is meant to have DC lighting to avoid this.  YS and MvC are investigating tonight as DC filter cabinet could well be faulty. (Subsequent to the meeting, the cabinet was found to be OK in principle, but not filtering the mains noise properly; Neuromag were informed.) CD informed the committee that the HPI was very variable in terms of efficiency. 
6. Feedback from Neuromag regarding shielded room.
Discussed in Matters Arising.
7. Lab Business and Progress

7.1  Sound insulation, progress – No real progress.
7.2  Acquisition software – Already discussed.
7.3  Simultaneous MEG and EEG recordings – There is a system for measuring impedances which was built by the Technical services and is working okay.  CD informed that there were some issues but Maarten and David are trying to address them.  
7.4  MEG Operator / Researcher Qualification Procedure – YS asked CD whether she thought Anna Shestakova (CSL) is ready to be left alone with the machine whilst operating.  CD reported that she was fine with the set up but hasn’t seen her do any tuning or other hardware-related tasks. RH and YS said it would be good for Anna to do the main operating for CSL who have a few block bookings coming up.  It was agreed to start with daytime hours only to start with in case she needs help from the CBU operators.  WM brought up the fact that some operators may be prepared to work after hours if paid over time.  YS suggested that we might allow a couple of extra slots per week.  WM pointed out that Elizabeth Fontenou will start as a CDF with him, so she might be able to be trained up as an operator as well.
7.5 Eye tracker – Still waiting for demo unit from the SMI.  WM queried what budget this would come out of, but YS informed that Elekta appear to be paying for this.  RH brought up that there as a possibility of discount if we had 2.  They would pay for an MEG one and we would pay for an MRI one.  WM will check the budget.
7.6 Visual supine stimulation – Maarten and Gary are investigating.  YS has found some MEG compatible goggles which he will look into.
7.7  Foot pedal – OH reported that this will be tested soon.  
7.8 Other issues – YS informed that we require more caps for EEG.  YS has got quotes and is expecting a demo from “Easycap” but had no response yet.  They are cheap and seem good quality.  Anthea informed YS that these will come under a different budget because they are each under £3,000.  
Action: YS to follow up “Easycap” demo.
8. GSK Update

YS and RH have short listed the applicants for the Post-doc to 4 people.  The interviews are on 3rd March and CD plus a GSK rep will be joining them on the interview panel.  There is one person who stands out from their experience and a number of other good candidates but they might not be able to start straight away as finishing their PhD.

RH informed that GSK have asked him about the possibilities of running another project starting at the end of 2008. In this case, a GSK project on schizophrenia would be in the focus. It would imply testing of 54 subjects – 27 of which would be control and 27 would be Schizophrenia patients.   After discussion it was agreed that it would be good for the CBU to allow this as MW pointed out, it creates a stronger relationship with the Industry.  However, FP asked whether the project would be a scientific collaboration with GSK or rather CBSU would be envisaged just to provide the facility for testing. He mentioned that the former is preferable, as providing research support for other institutions may draw upon resources the CBSU needs for successfully mastering its scientific research program. Discussion followed, and WM, YS and FP suggested that CBU researchers and GSK could aim at a scientific collaboration.  RH mentioned that he was not particularly interested in participating in the schizophrenia project with GSK. It was then agreed that FP will take a lead role in that and that RH will pass his details to Pradeep Nathan, the GSK main contact.
Action: RH to speak to GSK about CBU and GSK collaborating for their new project and to pass the GSK contacts onto FP.
9. External MEG software access
Discussed in item 4.  RH brought up the topic about out of hours testing with partner researchers.  He reported that GC said that they don’t normally allow non members of staff to record out of hours.  Also, ‘visiting workers’ don’t get access to the lab etc, but WM said that if they are willing to pay for the overtime of one of our operators to stay behind then that could work.  RH will look into.

10. MEG bookings for March
GM showed the committee a draft spreadsheet which shows how many slots researchers have used and how many they are waiting to receive.  It also shows the available slots in March and the amount each researcher has been allocated for that month.  This was done by prioritising each case and it worked out that each researcher received the same number of slots, also leaving one left each week in case of maintenance needs etc.  Please attach final slot distribution!
Action:  GM to amend/update spreadsheet with correct info regarding project codes etc.
11. Any other business (AOB) 
None.
12. Date of the next meetings (DOTNM)
Thursday 6th March @ 1pm in the WWSR.
